Author's response to reviews

Title: A postoperative neurological aggravation with sevoflurane in patient with Xeroderma Pigmentosum: a case report.

Authors:

salaheddine fjouji (fjoujisalah@gmail.com)
musatpha bensghir (mustaphabens_15rea@hotmail.com)
bahija yafat (yafa.bahija@hotmail.fr)
najib bouhabba (n.bouhabba@gmail.com)
lhoucine boutayeb (elhoucineboutayeb@hotmail.com)
hicham azendour (hzendour@gmail.com)
nordine drissi kamili (n.drissikamili@hotmail.com)

Version: 6 Date: 4 September 2012

Author's response to reviews:

in our case was probably due to sevoflurane. Indeed with propofol, muscle relaxants and opioids no postoperative neurological worsening was described with respect to the direct effects of these drugs.

by cons with sevoflurane, cases of experimental and clinical worsening were described in the presence of xeroderma pigmentosum